Author's response to reviews

Title: Efficacy of the epidural blood patch for the treatment of post lumbar puncture headache. BLOPP: A randomised, observer-blind, controlled clinical trial. [ISRCTN71598245]

Authors:

Regilio Oedit (r.oedit@erasmusmc.nl)
Fop van Kooten (f.vankooten@erasmusmc.nl)
Stef L Bakker (sbakker@amphia.nl)
Diederik W Dippel (d.dippel@erasmusmc.nl)

Version: 2 Date: 1 July 2005

Author’s response to reviews:

Dear Emma Parkin,

I have made all changes requested by you in the revised manuscript.

All unnecessary capitals were removed from the title.
All stops were removed from the title and subtitle.
The corresponding author is indicated on the title page including address and email address.
The date is removed from the title page.
The headers are removed from the manuscript.
Introduction is renamed into background.
References are (and were) cited in the text using consecutive numbers in square brackets.
References are placed before the full stop.
Conclusion and a list of abbreviations were added to the manuscript after the discussion.
The blank page (12) is removed.
The table was checked and only a minor change was needed.
The figure was cropped.

With best wishes,

Fop van Kooten